Author's response to reviews

Title: Inflammatory (B) symptoms are independent predictors of myelosuppression from chemotherapy in Non-Hodgkin Lymphoma (NHL) patients - analysis of data from a British National Lymphoma Institute phase III trial comparing CHOP to PMitCEBO.

Authors:

Rohini Sharma (r.sharma@imperial.ac.uk)
David C Cunningham (David.Cunningham@rmh.nhs.uk)
Paul Smith (p.smith@ctc.ucl.ac.uk)
Graham R Robertson (grobertson@med.usyd.edu.au)
Owen Dent (owen.dent@netspeed.com.au)
Stephen J Clarke (sclarke@med.usyd.edu.au)

Version: 4 Date: 12 May 2009

Author's response to reviews:

Editorial changes to format made as requested.